Author's response to reviews

Title: Beta hCG Secreting Non-Small Cell Lung Cancer: A relation spelling doom? A Case Report Saakshi Khattri MD1, Abhirami Vivekanandarajah MD2, Matthew Neimark MD3, Seema Varma MD4, Frank Kong MD5 1,2,3 Department of Internal Medicine, 4Department of Hematology Oncology, 5Department of Pathology, Staten Island University Hospital, New York 10305 CORRESPONDING AUTHOR Saakshi Khattri 475 Seaview Avenue Staten Island University Hospital Staten Island, NY 10305 Email- saakshi_doc@hotmail.com COAUTHORS Abhirami Vivekanandarajah Email- avivek27@gmail.com Matthew Neimark Email- Matthew_neimark@siuh.edu Seema Varma Email- seema_varma@siuh.edu Frank Kong Email- frank_kong@siuh.edu

Authors:

Saakshi Khattri (saakshi_doc@hotmail.com)
Abhirami Vivekanandarajah (avivek27@gmail.com)
Seema Varma (seema_varma@siuh.edu)
Frank Kong (frank_kong@siuh.edu)

Version: 2 Date: 30 June 2010

Author's response to reviews:

We have updated the abstract section as the format for submitting case reports expects. Secondly authors contributions have been specified.